Author's response to reviews

Title: Comparison of group counseling with individual counseling in the comprehension of informed consent: a randomized controlled trial

Authors:

Rajiv Sarkar (rsarkar@cmcvellore.ac.in)
Thuppal V Sowmyanarayanan (sowmyanarayanan@cmcvellore.ac.in)
Prasanna Samuel (prasanna_samuel@hotmail.com)
Azara S Singh (azarasingh@yahoo.com)
Anuradha Bose (abose@cmcvellore.ac.in)
Jayaprakash Muliyl (jayaprakash@cmcvellore.ac.in)
Gagandeep Kang (gkang@cmcvellore.ac.in)

Version: 4 Date: 27 April 2010

Author's response to reviews:

Dear Editor,

In response to our revision, only one comment was made. In this revision, we have replaced the sentence in the Discussion with the sentence suggested by the Associate Editor.

We hope that the article will be considered suitable for publication.

Yours sincerely,

Gagandeep Kang